Author's response to reviews

Title: Effects of CPAP on nitrate and norepinephrine in severe and mild/moderate sleep apnea

Authors:

Paula Pinto (paulagpinto@gmail.com)
Cristina Bárbara (cristina.barbara@chln.min-saude.pt)
Joseph M Montserrat (jmmontserrat@ub.edu)
Rita S Patarrão (rita.patarrao@fcm.unl.pt)
Maria P Guarino (maria.guarino@fcm.unl.pt)
Miguel M Carmo (mabmc@sapo.pt)
Maria P Macedo (paula.macedo@fcm.unl.pt)
Cristina Martinho (criscmartinho@netcabo.pt)
Ana R Dias (aaritaa@gmail.com)
Maria JM Gomes (marques.gomes@fcm.unl.pt)

Version: 5 Date: 21 January 2013

Author's response to reviews:

Dear Prof. Tonilynn Manibo

On behalf of the authors, I am pleased to re submit the article entitled “Effects of CPAP on nitrate and norepinephrine in severe and mild/moderate sleep apnea” with the trial registration number (ClinicalTrials.gov Identifier: NCT01769807) as the last line of the abstract, as requested.

We would like to thank to the editor and the reviewers by their comments and suggestions along the submission process.

Kind regards,

Paula Pinto